

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Daum, Thomas et al.

Article — Published Version Nutrition-sensitive lockdowns: Conceptual framework and empirical insights from Africa during COVID-19

Development Policy Review

Provided in Cooperation with: John Wiley & Sons

Suggested Citation: Daum, Thomas et al. (2023) : Nutrition-sensitive lockdowns: Conceptual framework and empirical insights from Africa during COVID-19, Development Policy Review, ISSN 1467-7679, Wiley, Hoboken, NJ, Vol. 41, Iss. 3, https://doi.org/10.1111/dpr.12666

This Version is available at: https://hdl.handle.net/10419/288046

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by-nc/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





ARTICLE

WILEY

Nutrition-sensitive lockdowns: Conceptual framework and empirical insights from Africa during COVID-19

Thomas Daum¹ | Hans Konrad Biesalski² | Nikola Blaschke¹ | Christine Bosch¹ | Denise Güttler¹ | Jakob Heni¹ | Juliet Kariuki¹ | Roseline Katusiime¹ | Anna Seidel¹ | Zinsou-Narcisse Senon³ | George Woode¹ | Regina Birner¹

¹Institute of Agricultural Sciences in the Tropics (Hans-Ruthenberg-Institute), University of Hohenheim, Germany

²Institute of Nutritional Sciences, University of Hohenheim, Germany

³Regional Agency of Agricultural Development- Plateau, Ministry of Agriculture, Livestock and Fishery, Republic of Benin

Correspondence

Thomas Daum, Institute of Agricultural Sciences in the Tropics (Hans-Ruthenberg-Institute), University of Hohenheim, Wollgrasweg 43, 70599 Stuttgart, Germany. Email: thomas.daum@uni-hohenheim.de

Funding information

Program of Accompanying Research for Agricultural Innovation

Summary

Motivation: Countries facing challenges of nutrition security confront a trade-off when dealing with pandemics such as COVID-19. Implementing lockdown measures, widely used worldwide, can help "flatten the curve" (of disease), but such measures may worsen nutrition security.

Purpose: We aim to identify and justify nutrition-sensitive lockdown measures to reduce trade-offs with nutrition security.

Methods and approach: We propose a conceptual framework which distinguishes eight lockdown measures and six pathways to nutrition security. To demonstrate the relevance of the pathways, we reviewed emerging literature on COVID-19 and nutrition security. We analysed the content of 1188 newspaper articles on lockdown effects in five African countries: Benin, Ghana, Kenya, Uganda and Zambia.

Findings: Some lockdown measures, such as closing workplaces and restricting movement, potentially worsen nutrition far more than others. Banning events and public gatherings have far less impact on nutrition. This can be

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes. © 2022 The Authors. *Development Policy Review* published by John Wiley & Sons Ltd on behalf of ODI.

seen from the framework, the academic literature, and is supported by the analysis of newspaper reports in the five countries.

Policy implications: It is better, when possible, to test and trace disease than to lock down. But when lockdowns are needed, the first recourse should be to measures that have few nutritional consequences, such as banning public events. When more drastic measures are necessary, nutritional harm should be mitigated by, for example, exempting farm labour from restrictions on movement, by replacing school meals with take-home rations, and, above all, providing income support to the most affected and most vulnerable households.

KEYWORDS

Africa, Covid-19, flatten-the-curve, food security, lockdown policies, nutrition

1 | INTRODUCTION

Confronted with COVID-19, governments across the world enacted lockdown policies to reduce contact between people. Lockdowns can comprise a variety of measures ranging from banning public events to workplace closures and stay-at-home orders (Hale et al., 2020; Hsiang et al., 2020). Lockdowns have been effective in slowing the contagion of the virus, preventing many deaths directly attributable to the virus (Hsiang et al., 2020; Lau et al., 2020), but they caused severe social and economic effects, which in turn have affected global health. The World Bank estimated that 97 million additional people fell into extreme poverty in 2020 (Mahler et al., 2021). The State of Food Security and Nutrition in the World annual report for 2020 (FAO et al., 2020) estimated that between 80 million and 130 million people became undernourished because of COVID-19 lockdowns and the associated global recession. While there are no global numbers, the emerging evidence indicates the significant effects of lockdowns not only on hunger but also on hidden hunger (micronutrient deficiencies) (Abay et al. 2020; Fore et al., 2020; Jaacks et al., 2021; Jafri et al., 2021; Kansiime et al., 2020; Sharma et al., 2020).

There is an emerging empirical literature on the overall effects of COVID-19 and the responses to the pandemic on nutrition security. Lockdowns have affected job opportunities, thus reducing household income and food consumption, even where food is available, as well as causing some supply chain disruptions (Abay et al., 2020; Adjognon et al., 2020; Amare et al., 2021; Jaacks et al., 2021; Kansiime et al., 2020; Mahmud & Riley, 2021). Lock-downs particularly affected poor urban households, which spend up to 70% of their income on food (Adjognon et al., 2020; Arndt et al., 2020; Laborde et al., 2020). Reduced food quantity particularly affects children. Headey et al. (2020) estimate that the pandemic caused a 14% rise in the prevalence of moderate or severe wasting among children. Roberton et al. (2020) estimate an increase of between 18% and 23% in child mortality.

Lockdowns also reduced access to, for example, micronutrients by affecting the consumption of fruits and vegetables and livestock products in low-income countries such as Ethiopia, Kenya, and South Africa (Abay et al., 2020; Kansiime et al., 2020) and among poor households in high-income countries such as the USA (Sharma et al., 2020).¹

¹In some population groups and world regions, lockdown policies also contributed to overweight by causing shifts to less healthy diets and lower physical activity levels (Poelman et al., 2021; Robinson et al., 2021).

WILEY 3 of 20

A lack of micronutrients is particularly harmful to vulnerable population groups, such as children. In the short term, it affects the immune system, which explains, among other things, the sharp increase in respiratory diseases in children during the pandemic (Sinha et al., 2020). In the long term, in particular for children during the 1000-day window from conception, it can cause irreversible effects on physical and cognitive development (Biesalski, 2017; Leddy et al., 2020). While disruptions to the access to calories typically become visible, e.g., in the form of empty food shelves and food riots during the food price crisis of 2007–2008 (Berazneva & Lee, 2013), and are likely to be corrected in democratic countries (Sen, 1981), the effects on the access to micronutrients are typically hidden and are, therefore, more likely to remain unaddressed.

The emerging literature quoted above has established important effects of the overall responses to COVID-19 on nutrition security. However, none of the studies has, so far, systematically disentangled the effects of different lockdown measures on nutrition security. This is important because lockdown policies comprise a wide variety of measures, ranging from stay-at-home orders to the closure of international borders. Such diverse measures may affect nutrition security in different ways and to different extents. To address this knowledge gap, this article develops a conceptual framework that can be used to examine how different lockdown measures affect different dimensions of nutrition security. The framework identifies potential pathways, by which different lockdown measures affect nutrition security. On this basis, "nutrition-sensitive lockdown policies" can be identified, which are effective from an epidemiological perspective but counter the negative trade-offs with nutrition security.

To assess the relevance of the pathways identified in the conceptual framework, we draw on two types of empirical evidence. First, drawing on statistical data on pre-COVID-19 nutrition security and COVID-19 lockdown stringencies, we explore whether more food-insecure countries implemented different lockdown measures as compared to more food-secure countries. Second, we draw on a content analysis of 1188 newspaper articles on lockdown effects that were published by major newspapers in five African countries: Zambia, Benin, Ghana, Uganda, and Kenya. The evidence from newspaper articles has to be interpreted carefully. Newspaper articles may not show the actual situation and nutrition security of a country, as they may be influenced by the degree of political freedom. As such, newspaper articles cannot replace evidence from scientific research. However, only a few thorough empirical studies could be conducted in developing countries *during* the first wave of lockdown measures. Such phone-based studies provide very valuable insights on nutrition security, but also have limitations (Hirvonen et al., 2021). Most importantly, these studies focused on nutrition security effects rather than the underlying mechanisms causing them. Therefore, newspaper articles appear as useful to assess the relevance of our conceptual framework, as they provide unique and real-time insights into the different ways that different lockdown measures have affected different dimensions of nutrition security.

As further detailed in the empirical methods section, the five case study countries were chosen to cover different levels of nutrition security at the onset of the pandemic, different types of political regimes, and different combinations and stringencies of lockdown measures.

Section 2 introduces our framework. Section 3 introduces the five case study countries and presents the methods used to understand the stringency of lockdown measures (see subsection 3.1.) and the effects of lockdown, namely the media analysis (see subsection 3.2.). Section 4 presents evidence from statistical data and newspaper articles. Section 5 discusses policy implications.

2 | FRAMEWORK

Table 1 presents the framework that was developed to identify potential pathways, by which different lockdown measures influence nutrition security. To differentiate the lockdown measures, we use the classification of the Oxford COVID-19 Government Response Tracker, which distinguishes eight types of lockdown measure (Hale et al., 2020), as shown in Table 1. To identify the pathways by which these measures may affect nutrition security, we draw on Devereux et al. (2020), who conceptualize potential impact pathways by combining three different frameworks: the Food and Agriculture Organization of the United Nations' (FAO) "four pillars" framework of food security, which

	Dimensions of nutritio	in security / Pathways of i	impact on nutrition security				
	Availability		Access				Total
Lockdown measures	Upstream: Agricultural production	Downstream: Processing, sales, and trade		Physical access	Transfers	Utilization	of nutrition security affected
(1) Workplace closing	Limited (if agriculture is exempted)	Limited (if food-related activities are exempted)	Strong (can affect incomes and savings/assets)	Limited	Strong (can affect private transfers)	Strong (loss of access can affect diets)	3/6
(2) Close public transport	Strong (can affect Iabour)	Strong (can affect Iabour)	Strong (can affect incomes and savings/assets)	Strong (can affect access to markets)	Strong (can affect private/public transfers)	Strong (can affect diets and access to health care)	5/6
(3) Restrictions on internal movement	Strong (unless agricultural labourers can be exempted)	Strong (unless agricultural labourers can be exempted)	Strong (can affect incomes and savings/assets)	Strong (can affect access to markets)	Strong (can affect private/public transfers)	Strong (can affect diets and access to health care)	6/6
(4) Stay-at-home requirements	Strong (unless agricultural labourers can be exempted)	Strong (unless agricultural labourers can be exempted)	Strong (can affect incomes and savings/assets)	Strong (can affect access to markets)	Strong (can affect private/public transfers)	Strong (can affect diets and access to health care)	6/6
(5) Restrictions on private gathering	Limited	Limited	Limited	Limited	Strong (can affect private transfers)	Limited	1/6
(6) School closure	Limited	Limited	Limited (but long-term effects on physical and mental development possible)	Limited	Strong (can affect school feeding programmes, but such effects can be avoided)	Strong (can affect diets)	2/6
(7) Cancel public events	Limited	Limited	Limited	Limited	Limited	Limited	0/6
(8) Restrictions international movement	Limited	Limited	Limited	Limited	Limited	Limited	9/0
Lockdown measures affecting nutrition security dimension	3/8	3/8	4/8	3/8	5/8	5/8	

TABLE 1 Potential effects of COVID-19 lockdown measures on nutrition security

Source: Authors.

DAUM ET AL.

WILEY

5 of 20

WILFY

distinguishes food availability, access, utilization, and stability (FAO, 2008); the food systems approach proposed by the UN Committee on World Food Security (HLPE, 2017); and Amartya Sen's (1981) entitlement approach.

FAO's four-pillar approach is useful to disentangle which dimension(s) of nutrition security a specific lockdown measure is likely to influence: (1) availability; (2) economic, social, and physical access; (3) utilization; and/or (4) the stability of the first three pillars over time (FAO, 2008). To assess the impact of different lockdown measures in more detail, it appears useful to split food availability into two categories: (1) agricultural production, which depends on the access to inputs such as fertilizer and labour as well as services such as finance, extension services, and veterinarians; and (2) food processing, sales, and international and national trade. Separating the upstream and downstream agri-food value chain helps provide a better understanding of entry points for supply disruptions, an aspect that is emphasized in food systems thinking (Devereux et al., 2020; Harris et al., 2020). Concerning FAO's access dimension, it appears useful to differentiate between economic access to food, which depends on the income and the price of food as well as savings and assets, and physical access to food can also be based on "transfer entitlements," which can be private (e.g., remittances or food donations) or public (e.g., public cash transfers and food aid). They are listed as a separate column in Table 1. Sen's "own-production entitlements," which are relevant for smallholder farmers, are captured under the column agricultural production.

FAO's utilization dimension depends on various aspects, such as access to health, sanitation, food safety, and dietary quality. This dimension is highly relevant for our framework, since lockdown measures may cause shocks that undermine food utilization. For example, certain lockdown measures lead to declining access to health precautions and medical treatment for childhood diarrhoea, which particularly affects vulnerable groups (Biesalski et al., 2016; Leddy et al., 2020).

Stability is treated as a cross-cutting dimension in our conceptual framework and is, therefore, not represented by a separate column in Table 1. The dimension of stability is related to resilience, a term that conceptualizes how well different systems can absorb, adapt, and recover from shocks (Ansah et al., 2019; Béné et al., 2016; Pingali et al., 2005). Nutrition security resilience has both a short-term and long-term perspective (Béné et al., 2016). This is relevant in our context since both perspectives have to be considered when assessing the nutrition sensitivity of lockdown measures.

The columns in Table 1 display the dimensions of nutrition security identified above. They can be interpreted as pathways, by which specific lockdown measures affect nutrition security. The cells in Table 1 show a qualitative assessment of the expected effect of a lockdown measure on the respective pathway. The assessment is based on theoretical considerations and the available literature, as further detailed below.

We distinguish two broad categories of potential impact: "limited" and "strong." For effects that are not obvious, an explanation has been added in italics in brackets. Some measures are nutrition-sensitive by their nature: they do not affect any of the pathways in a significant way. Other measures can be made nutrition-sensitive by making exemptions or implementing countervailing measures. For example, school closures can affect access to school meals, but if alternative ways to distribute such school meals are established, their effect on nutrition security diminishes. Also, restrictions of internal movements have strong effects, though such effects can be reduced—in principle—if agricultural labourers are exempted, which can be difficult as countries may not be able to control and distinguish between agricultural and non-agricultural workers.

The final column in Table 1 presents a score, which indicates the number of dimensions for which a strong effect is expected. This score can be seen as an indication of the extent to which a specific lockdown measure affects nutrition security: the higher the score, the less nutrition-sensitive a measure—if no specific provisions are made to counter the effect. However, the score only reflects the number of pathways affected. The overall effect of a measure on nutrition security will depend on how large the effect is within each pathway. The last row presents a score that shows how many of the eight lockdown measures are likely to affect a particular dimension of nutrition security. This score, thus, reflects the pathways, which are expected to be most relevant for negative effects on nutrition security, if all lockdown measures were applied jointly.

Workplace closures could in principle have a very large effect on upstream and downstream food availability. Anticipating this problem, countries have typically made exemptions for farm production and food value chains. Therefore, Table 1 indicates that workplace closures are expected to have limited direct effects on food availability, as long as essential workplaces are exempted.

This was not always done during COVID-19. For example, wet markets were often closed, disrupting food supply chains and physical access to food (Pereira & Oliveira, 2020). Moreover, indirect effects are possible. For example, closing banks may affect farmers' access to credit for seed and fertilizer (Devereux et al., 2020). Workplace closures affect wage labour—i.e., for workers in the informal sector—in which jobs often require physical attendance and are not covered by unemployment schemes (Amare et al., 2021; Arndt et al., 2020; Barrett, 2020; Devereux et al., 2020; Jaacks et al., 2021; Mahmud & Riley, 2021; Wouterse et al., 2020).

Income drops may be mitigated by using savings or selling assets—undermining future resilience—but many households do not have such buffers (Laborde et al., 2020; Mahmud & Riley, 2021). Where private food and cash transfers are common, losing jobs does not only affect the breadwinners' households but also the people who receive food and cash from these households. Falling food demand may cause price declines, affecting farmers' revenues and future investments (Harris et al., 2020). In some cases, workplace closures may also have contributed to increasing agricultural production, for instance where migrant workers returned to rural areas and started to work on the farms.

Closure of public transport and restrictions of internal movements have similar effects, but public transport closures do not affect the mobility of people who have private means of transportation (unless private vehicles are included in the restrictions). However, they can disrupt agricultural production and the food supply chain, for example, by affecting the movements of agricultural workers as well as inputs and outputs (Ayanlade & Radeny, 2020; Ceballos et al., 2020; Hammond et al., 2022; Huber et al., 2018; Minten et al., 2020; Wouterse et al., 2020). Movement restrictions are likely to particularly affect the production of nutritious but perishable food such as dairy and horticulture products (Harris et al., 2020; Kansiime et al., 2020, Minten et al., 2020; Torero, 2020). Movement restrictions are likely to slow down the overall economy, thereby affecting own-labour entitlements, and may also undermine physical access to food (Hammond et al., 2022; Kansiime et al., 2020) and private transfers, as social networks can become disrupted at a crucial moment (Devereux et al., 2020). They can also affect access to health care, thus undermining food utilization.

Stay-at-home requirements are an extreme lockdown measure that affects not only the economic and physical access to food but also private and public transfer entitlements. Stay-at-home requirements particularly affect the physical access to non-storable and perishable food, which can inter alia affect food utilization by reducing dietary diversity (Harris et al., 2020). School closures can affect food transfer programmes, which are often organized via schools, thereby affecting children, who are among the most vulnerable to food and nutrition shocks (Ali et al., 2020; Laborde et al., 2020). In principle, alternative ways to distribute school meals during school closures can be used, such as take-home rations, vouchers, and cash transfers. However, targeting problems may occur. Prolonged school closures can affect educational achievements and, subsequently, job opportunities, inter alia.

Some measures are likely to have limited effects on nutrition security, particularly **restrictions on private gatherings**, **cancellation of public events**, and **international movement of people**. This is not to say that they have no effects. For example, **restricting private gatherings** may affect private transfer entitlements, to use Sen's terminology. Social events in rural areas, such as weddings and funerals, may provide poor people in a village with access to high-quality food, e.g., meat that they cannot afford otherwise (Jahnke, 1982, p. 74). **Cancellation of public events**, such as banning public church services, may affect the ability of food/cash programmes run by faith-based organizations to raise funds. Also, group-based extension delivery to farmers may be affected. **Restrictions on international movements** have disrupted supply chains for horticultural products, such as cut flowers and fruits that are typically transported in passenger planes (Laborde et al., 2020). Border control measures may lead to disruptions, even where agricultural and food commodities are exempted (Torero, 2020). **Restrictions on international movements** may also affect economic access for families relying on incomes from international migrant labourers.

Table 1 shows that the lockdown measures affecting most nutrition security dimensions are related to mobility and workplace closures. The closure of public workplaces affects five of eight dimensions, and restrictions on internal

WILEY 7 of 20

movement and stay-at-home requirements affect four of eight dimensions each. Workplace closures affect three of eight dimensions. As the last row in Table 1 shows, the largest share of lockdown measures (five of eight) reduces nutrition security by affecting transfer entitlements and utilization. The next most important pathway is the impact on economic access, as four of the eight lockdown measures are likely to reduce economic access. Since lockdown measures that have a potentially strong impact on food availability (workplace closing and restriction on internal movement) were typically implemented with exemptions for agriculture and food chains, this impact pathway plays a relatively limited role as compared to the other pathways.

3 | EMPIRICAL METHODS

Two types of empirical evidence are used to assess the relevance of the impact pathways that were identified in the conceptual framework above. The first is a comparison of the stringency of different lockdown measures implemented by countries that differ in their food security status (Section 3.1). The second is an analysis of newspaper articles from five selected African countries (Section 3.2). The methods used to provide these two types of empirical evidence are described in more detail in the following.

3.1 | Stringency of lockdown measures in countries that differ by food security status

The calculation of the stringency of the different lockdown measures is based on data from the Oxford COVID-19 Government Response Tracker (Hale et al., 2020). The Tracker captures the government responses to the COVID-19 pandemic in 186 countries. Its lockdown dimensions comprise eight different measures, as shown in Table 1 above. The Tracker distinguishes three to four different levels of stringency for each measure on an ordinal scale. Aside from the lockdown measures on international movement, all restrictions have "flags," which indicate whether the measures are only applied in targeted regions or to the whole population. The stringency of each of the eight lockdown measures between January 2, 2020, and May 17, 2020, for each of the case study countries, was calculated as follows: the daily lockdown scores (points according to the ordinal categorization) for each of the measures were aggregated and rescaled by their maximum possible scores to create an index between 0 and 100.

In addition to exploring the lockdown stringency of the five case study countries, Section 4 also compares the lockdown stringencies between countries with low and high food security status. For this purpose, countries with a prevalence of undernutrition above 10% (World Bank, n.d.), using data from 2018, or a severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2016–2018, were classified as having low food security (low FS). Countries with a prevalence of undernutrition below 10% (World Bank, null), using data from 2018, and no severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2018, and no severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2018, and no severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2018, and no severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2018, and no severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2016–2018, were classified as having high food security (high FS).

3.2 | Content analysis of newspaper articles in five African countries

We draw on a content analysis of newspaper articles that deal with nutrition security effects of lockdown measures in five African countries to provide further evidence on the relevance of the different impact pathways identified by our conceptual framework: Zambia, Ghana, Benin, Kenya, and Uganda. These countries were chosen to reflect different levels of nutrition security before the outbreak of COVID-19—ranging from moderate in Ghana to serious in Benin, Uganda, and Kenya and alarming in Zambia—and different levels of political "freedom" (Freedom House, 2020), which can influence the incentives of governments to ensure nutrition security (Blaydes & Kayser, 2011; Sen, 1981, 2001; Thomson, 2019). Sen (2001) famously argued that, "no famine has ever taken place in the history of the world in

8 of 20

a functioning democracy" because governments "have to win elections and face public criticism, and have strong incentive to undertake measures to avert famines and other catastrophes."

The countries also have different government capacities as measured by the Quality of Government Index (see Table 2).

In each country, one of the largest public² and a main private newspaper were chosen: *The Lusaka Times* and *The Mast* in Zambia, *The Daily Graphic* and *My Joy Online* in Ghana, *La Nation* and *La Nouvelle Tribune* in Benin, *The Daily Nation* and *The Standard* in Kenya, and *The New Vision* and *The Daily Monitor* in Uganda. Due to challenges with the search function of some of the newspaper's online archives, articles were then identified with Google using the search terms "covid lock" and "covid cordon" (in Benin). The period of analysis was between January 1, 2020, and May 17, 2020, when the lockdown measures in most countries were eased again. Newspaper articles merely covering case numbers were excluded.

The remaining 1188 articles (in Zambia: 136; in Ghana: 437; in Benin: 81; in Kenya: 231; in Uganda: 304) were coded using the Qualitative Data Analysis (QDA) Miner software. Newspaper articles were pre-coded by five people on the author team using a jointly developed coding structure mirroring the dimensions of nutrition security developed in Section 2 and then coded by the lead author to ensure consistency. Only actual, observed food security effects stemming from the lockdown policies in the respective country were coded: theoretical warnings of possible effects were not coded and lockdown effects reported from other countries were not considered. Articles that discussed different food security effects of lockdown policies received multiple coding.

4 | EMPIRICAL EVIDENCE

4.1 | Evidence from statistical data

Figure 1 compares the lockdown response for countries with low and high levels of food security before the pandemic. The figure shows that countries with a lower level of food security are, on average, more likely to refrain from measures that have large effects on nutrition security, such as workplace closures and stay-at-home orders, or to enact them over a shorter period. Analysis of variance (ANOVA) and pairwise Tukey tests show that differences are significant for internal movement restrictions (10% level), school closures (5%), cancellations of public events, workplace closures, and stay-at-home requirements (1% level). Figure 1 also shows that countries—regardless of their food security status—pursued higher lockdown stringencies for measures that are less invasive and affect fewer dimensions of nutrition security, such as restrictions on private gatherings, school closures, cancellations of public events, and restrictions on international movement.

4.2 | Evidence from newspaper articles in five African countries

In this section, the effects of lockdowns on nutrition security in five African countries—Zambia, Benin, Ghana, Uganda, and Kenya—are explored empirically. Table 3 shows the COVID-19 lockdown strategies of the five case study countries, revealing how lockdown decisions have varied in countries characterized by low food security (which were grouped in Figure 1). Zambia has never restricted the movement of people (lockdown measures 2, 3, 4), but did close schools (lockdown measure 6) and entertainment workplaces (lockdown measure 1). A stricter lockdown was ruled out because of food security concerns, with the president noting: "I am aware that some of you have been saying, 'We would rather die from Covid-19 than from hunger." (Cara, 2020).

²In Africa, public newspaper ownership is a widespread phenomenon and such newspapers are among the most widely read in Africa (Moehler & Singh, 2011).

TABLE 2 Case study country characteristics

	Zambia	Benin	Ghana	Uganda	Kenya
Country characteristics					
Global Freedom Index (Freedom House, 2020)	54 (partly free)	66 (partly free)	82 (free)	34 (not free)	48 (partly free)
Global Hunger Index 2019 (von Grebmer et al., 2019)	38 (alarming)	24 (serious)	14 (moderate)	31 (serious)	25 (serious)
ICRG Indicator of Quality of Government 2020 (PRS, 2022)	0,42	NA	0,51	0,44	0,51
Newspaper/article selection					
Selected newspapers	The Lusaka Times, The Mast	La Nation, La Nouvelle Tribune	The Daily Graphic, My Joy Online	The New Vision, The Daily Monitor	The Daily Nation, The Standard
Number of articles (n)	136	81	437	304	231

Notes: The Global Freedom Index ranges from 0–100, with 100 indicating the highest degree of freedom. The Global Hunger Index ranges from 0–100, with values above 10 indicating moderate levels, above 20 indicating serious, and above 35 alarming levels. The Quality of Government Index ranges from 0 (lowest) to 1 (highest) and covers the variables "Corruption," "Law and Order," and "Bureaucracy Quality."

Benin established a cordon sanitaire—a quarantine zone—around the major cities for three weeks. People were allowed to move freely within and outside the zone but were not allowed to cross the borders of this zone. Food security concerns influenced Benin's lockdown, with the president saying: "Who can wait two, three, or four weeks even without working and living on monthly income?" (Vidjingninou, 2020).

Ghana followed a more stringent approach but only in its metropolitan areas, where stay-at-home requirements (lockdown measure 4) were enacted and non-essential businesses had to close (lockdown measure 1). Many measures were lifted after three weeks because of the "severe" nutrition security impacts on the poor and vulnerable (Adebayo, 2020).

Uganda enacted a hard lockdown, banning public and private transportation (lockdown measure 2), public gatherings (lockdown measure 7), closing all non-essential workplaces (lockdown measure 1), and closing schools (lockdown measure 6), and applying a dusk-to-dawn curfew (lockdown measure 4).

Kenya applied a nationwide dusk-to-dawn curfew similar to that of Uganda, but Kenya was less strict regarding the closure of businesses (focusing only on entertainment workplaces) and public transportation (only restricting the number of passengers).

Table 4 shows how the COVID-19 lockdown strategies of the five case study countries translated into the stringency assessments of the Oxford COVID-19 Government Response Tracker (between January and mid-May 2020). Uganda and Kenya were the most stringent of the five countries. On average, the five countries chose lockdown measures with fewer implications on nutrition security, such as restrictions of private gatherings (affecting one of eight nutrition security dimensions), school closures (affecting two of eight nutrition security dimensions), cancelling public events (affecting none of the eight nutrition security dimensions), and restrictions on international movement (affecting none of the eight nutrition security dimensions) (as set out in Table 1). However, Uganda and Kenya, in particular, enacted lockdown measures with potentially severe nutrition security implications, such as workplace closures (affecting three of eight nutrition security dimensions, see Table 1) and stay-at-home orders (affecting four of eight nutrition security dimensions).

WILFY^{__}

Birner et al. (2021) consider the political economy of the lockdown decisions of the five African countries, suggesting that various factors explain lockdown stringencies, including political freedom, the expected scale of opposition, the capacity to offset adverse effects, the willingness to use state force to ensure adherence, food security considerations, and policy diffusion.

Table 5 shows how lockdowns have affected nutrition security in the five case study countries, based on a content analysis of articles published by the major newspapers in these countries (see section 3.2). Table 5 suggests that nutrition security effects occurred in all countries, but that they were most prevalent in countries with stricter lockdowns, such as Uganda and Kenya.

Across all countries, economic access to food was the most frequently mentioned nutrition security dimension, affected by four of the eight lockdown measures according to the conceptual considerations (see Table 1). Economic access was particularly undermined in Uganda, which imposed the most far-reaching workplace closures (lockdown measure 1), followed by Ghana and Kenya, which also enacted workplace closures but for a shorter period (in Ghana) and affecting fewer workplaces (in Kenya). The effects of workplace closures on economic access are illustrated by the following two quotes, of which the latter highlights the long-term effects of prolonged lockdowns:



FIGURE 1 Stringencies of different lockdown types, comparing high and low food security (FS) countries. *Notes*: Lockdown stringencies between January 1, 2020, and May 17, 2020, based on data from the Oxford COVID-19 Government Response Tracker (Hale et al., 2020) (see section 3.1). Countries with a prevalence of undernutrition above 10% (World Bank, null), using data from 2018, or a severe/alarming food security status according to the World Hunger Index 2019 (von Grebmer et al., 2019), using data from 2016–2018, were classified as having low FS. Levene's test shows that the variances for each group are equal. The boxplots show the median stringency with the whiskers indicating the 25th and 75th percentile. Asterisks indicate significant differences at the 10% (*), 5% (**), and 1% (***) levels. Sample size: 176 countries.

Zambia Benin Type Ghana Uganda Kenva (1) Entertainment Entertainment Non-essential Non-essential Entertainment Workplace workplaces workplaces workplaces in workplaces workplaces closing (30 days*) within cordon major cities (42 days[#]) (55 days) sanitaire (21 days) (49 days) Ban of (mini) (2) No In major cities Yes Reduced capacity Close public buses (21 days) (54 days) (59 days) transport (49 days) (3) No Cordon sanitaire In major cities Yes Yes Restrictions (43 days) (21 days) (50 days) (52 days) on internal movement (4) No No In major cities Dusk-to-Dawn-Dusk-to-Dawn-Stay-at-home Curfew Curfew (21 days) requirements (50 days) (52 days) (5) Max. 50 people Max. 10 people Max. 5 people Yes Yes Restrictions (53 days) within cordon (63 days) (50 days) (66 days) sanitaire on private gatherings (43 days) (6) Ves Yes Ves Ves Ves School closure (59 days) (43 days) (63 days) (59 days) (64 days) Max. 50 people (7) Yes Yes Yes Yes Cancel public (42 days) (62 days) (63 days) (61 days) (66 days) events (8) Screening/ Screening/ Ban of Screening/ Ban of Restrictions on quarantine quarantine non-nationals/ quarantine non-nationals/ international (89 days) (59 days) residents (119 days) residents movement (63 days) (62 days)

 TABLE 3
 Lengths of lockdown measures in case study countries between January 1 and May 17

Notes: *Except bars and taverns. # Gradual reopening. Days were calculated from January 1, 2020, to May 17, 2020. *Source*: Authors.

"There has been loss of lives and devastating blows to economies and, particularly, livelihoods of citizens who live from hand to mouth" (Onyango-Obbo, 2020).

"I have a family of three children and before receiving relief I was using my savings to buy food every day. Later, the money was exhausted when the President extended the lockdown for the extra 21 days" (Covid-19: How residents of Kabarole, 2020).

Stay-at-home orders (lockdown measure 4), which were enacted in parts of Zambia and Ghana and across Uganda and Kenya, can similarly undermine economic access to food (unless traveling to workplaces is specifically exempted). As shown in the conceptual framework, such a measure can potentially affect four of eight nutrition security dimensions. Unlike the closure of formal workplaces, such measures also affect informal workplaces. The effects of stay-at-home orders on economic access are illustrated by the following quote:

"Vulnerable homes whose livelihoods depend on hand to mouth small scale businesses (...) have now reached their lowest ebb due to, among others, the stay-at-home preventive measure" (Lockdown has triggered rise in food prices, 2020).

WILEY-

ockdown	Workplace	Closing public	Restrictions of internal	Stay-at-home	Restrictions of private		Cancel public	Restrictions of international	
neasures	closing	transport	movement	requirements	gatherings	School closure	events	movement	Average
Zambia	11	0	21	11	33	46	12	48	23
Benin	19	26	22	0	31	37	48	35	27
Shana	23	26	10	12	39	49	49	58	33
Jganda	40	42	45	29	38	46	48	54	43
ćenya	36	31	41	30	52	50	52	60	44
iverage	26	25	28	16	39	46	42	51	
nte: Between Jan	uary 1, 2020, and	May 17, 2020. Cold	ours are based on c	onditional formatti	ing of the unround	ed values, ranging t	from dark green (0)	to dark red (60).	

TABLE 4 Lockdown stringencies in the case study countries.

2 j0 ມ -0 0 Source: Oxford COVID-19 Government Response Tracker. IVIA J IVIA V 2 Note: Be

WILEY

Dimensions of food and nutrition security	Agricultural production	Food processing, sales, and trade	Economic access	Physical access	Transfers	Utilization	Average
Zambia (n = 136)	1	4	7	3	2	2	3
Benin (n = 81)	0	0	5	1	0	0	1
Ghana (n = 437)	1	2	13	6	1	0	4
Uganda (n = 304)	5	10	23	7	2	2	8
Kenya (n = 231)	1	6	11	4	2	0	4
Average	2	4	12	4	1	1	

TABLE 5	Share (%) of newspa	per articles	reporting food	and nutrition	security effects.
		/				

Note: Between January 1, 2020, and May 17, 2020. Colours are based on conditional formatting of the unrounded values, ranging from dark green (0) to dark red (23).

Source: Authors.

Economic access was also affected by the mere anticipation of stay-at-home orders. This led to panic buying in all the countries and, subsequent price spikes, which undermine the economic access of poorer people to food, as the following quote from Ghana illustrates:

"Amid the scramble for essential foodstuffs and products, price gouging and hoarding has been the order of the day" (Ayitey, 2020).

Stay-at-home orders are an extreme form of movement restriction. As with more mild forms of movement restriction, such as restrictions of internal movements and closing public transport (lockdown measures 2 and 3), which can potentially affect five of eight nutrition security dimensions according to the conceptual considerations, such measures can undermine the physical access to food and destroy private social safety nets, as the following quotes illustrate:

"Our markets are mostly far from our residential areas especially in Accra and Kumasi where the lockdown is biting" (Bedzrah, 2020).

"Lockdowns are also very difficult for people...whose survival is provided by friends and relatives through our African 'Ubuntu' culture" (Mwansa, 2020).

Restrictions of the internal movement of people and closing public transport systems also affected agricultural production, food processing, and sales, as the following quotes illustrate:

"The restrictions on movement of vehicles are disrupting the supply-chain [of poultry] at various levels – hatcheries, feed mills, poultry farms and retail outlets. The fact that there is mortality and perishability involved in the trade makes it vulnerable" (Covid-19: How a poultry farmer, 2020)

"With the ban of public transport, the food prices will increase because most of the vendors buy food from farmers who have been transporting it" (Vendors opt to sleep, 2020).

In principle, the negative effects of lockdowns on nutrition security, in particular the loss of economic access to food following workplace closures, can be mitigated with cash transfers and food aid, as long as governments can afford such programmes, and food is available. The case study countries that enacted strict workplace closures—Ghana, Uganda, and Kenya—set up such programmes. However, these suffered from various political economy and governance challenges. In Uganda, public food aid was misused to ensure votes for the imminent presidential elections, and

/ILEY-

providing food aid outside the official government channels was forbidden (Anguyo, 2020). The politicization of food aid was particularly frequently reported in Uganda, but also appeared as a problem in Kenya and Ghana:

"A woman has further lamented her ordeal of being denied food at Shiashie because she was not a card bearing member of the ruling NPP....This action if not curtailed with immediate effect will lead to satisfying party supporters rather than the targeted vulnerable women and children" (Politicisation of food sharing, 2020).

5 | POLICY IMPLICATIONS

Wiery

Drawing on conceptual considerations, the emerging empirical literature, and a content analysis of Covid-19 effects in 1188 articles from 10 major newspapers in five African countries, this article has explored the effects of different lockdown measures on different dimensions of nutrition security. Disentangling such effects can help policy-makers design "nutrition-sensitive lockdown policies." The conceptual framework has shown that some lockdown measures, such as workplace closures and restrictions of movement, have far-reaching consequences on nutrition security, affecting several nutrition security dimensions, while others, such as banning public events, have limited effects. As such, measures such as the latter should be prioritized.

The framework suggests lockdowns particularly affect economic access to food—a nutrition security dimension that is undermined by workplace closures and restrictions of movement implemented in all case study countries. This hypothesized decline in economic access was confirmed by the content analysis—across the five countries, 12% of the coded newspaper articles mentioned this pathway. The emerging empirical research underlines the relevance of this pathway as well (Amare et al., 2021; Arndt et al., 2020; Devereux et al., 2020; Kansiime et al., 2020; Laborde et al., 2020).

While the conceptual framework has suggested that several lockdown measures undermine food transfers (which can be private and public), this pathway was not covered by the newspaper articles. Similarly, the newspapers did not frequently cover lockdown effects on food utilization, which our framework identified as a key nutrition security dimension that was affected. However, this may be because the dimension of food utilization is less tangible than other dimensions. For example, in contrast to declining access to calories, declining access to micronutrients is hidden—both to citizens and journalists—and the most affected population groups, such as women and children, may have less voice in public debates. Yet, while households may continue to access enough calories, dietary shifts towards cheaper, less nutritious diets may affect health and, for children, also mental and physical development, affecting them long after lockdowns are lifted (Kansiime et al., 2020; Leddy et al., 2020).

Both conceptual and empirical findings suggest the access pathway is more important than the food availability pathway. Therefore, guaranteeing access to food should be a top priority for policy-makers. Few newspaper articles reported challenges related to agricultural production at the time of the study, potentially because of widespread exemptions (e.g., allowing the free movements of agricultural labourers). However, some challenges may also become visible only after some delay and may have not been fully captured by the newspaper articles. Hammond et al. (2022) report a wide range of challenges that African smallholder farmers faced concerning COVID-19-related restrictions. Challenges related to downstream value chain steps such as processing, sales, and trade were reported more frequently by the newspapers, a trend that confirms the literature (Harris et al., 2020; Kansiime et al., 2020; Minten et al., 2020; Torero, 2020). However, overall, availability challenges were less commonly reported than economic access. The early emphasis on availability in the international debate on COVID-19 and nutrition security may have helped to ensure that agricultural inputs and outputs and food kept moving in the first phases of Covid-19. However, this emphasis may have drawn attention away from the important questions on how to ensure that citizens can keep accessing food in the first place.

Understanding the effects of different lockdown measures on nutrition security can help policy-makers design nutrition-sensitive lockdowns measures during COVID-19 and with future pandemics in mind. Policy-makers usually struggle to address multiple concerns at any given moment, and hence need to prioritize. Therefore, the following section identifies priorities based on the theoretical framework and the empirical findings:

14 of 20

- Test-and-trace strategies should be prioritized by policy-makers, as long as they are sufficient from a containment
 perspective and as long as capacities for such a strategy exist, as lockdown measures can have implications on
 nutrition security.
- If lockdown measures become necessary, countries should prioritize measures that have limited effects on nutrition security. For example, banning large public events has little effect on nutrition security.
- 3. If such measures are not sufficient, countries can enact measures which potentially damage nutrition security, but whose damage can be reduced with relatively simple counter-measures. For example, movement restrictions can reduce farm workers' mobility, thus undermining agricultural production and food availability, but such negative effects can be avoided by granting exemptions for anyone involved in farming. Similarly, school closures can have significant implications for nutrition security, but negative effects can be reduced by providing take-home rations when schools are closed.
- 4. Measures damaging to nutrition security and whose effects cannot (easily) be mitigated by means of counter-measures—or whose timely implementation is not feasible for some countries—should be avoided as far as possible unless the direct health effects of pandemics and endemics outweigh any trade-offs. For example, workplace closures and stay-at-home orders slow contagion, but they can heavily undermine the economic and physical access to food. In the case study countries that have relied on such measures—in particular Uganda and Kenya—negative nutrition security effects were more frequently reported, confirming the findings of Kansiime et al. (2020, p. 2), who found that these measures had "significant ramifications on food security." Whenever lockdown measures that can potentially undermine nutrition security have to be chosen, putting in place strong measures to guarantee the availability and access to food has to be a top priority for policy-makers.

In principle, cash transfers and food aid programmes can buffer the nutrition security effects of such lockdown measures (Gerard et al., 2020; Wouterse et al., 2020). In Ethiopia, for example, households participating in the Productive Safety Net Programme experienced only modest negative nutrition security effects (Abay et al., 2020). India is another example where such programmes played a big role to buffer the nutrition shocks experienced due to the economic slowdown and lockdown policies related to COVID-19 (Sinha, 2021).

However, merely accompanying otherwise "nutrition-insensitive" lockdowns with such social protection programmes may be both a costly and a dangerous strategy—in particular in countries with poor governance capabilities. All the case study countries relying on food transfers—Ghana, Kenya, and Uganda—faced governance challenges, such as targeting problems and political patronage, albeit to different degrees (see also Birner et al., 2021). In Uganda, for example, public food aid was misused for political patronage due to the imminent presidential elections, and food aid provided outside the official government channels was exposed to the charge of "attempted murder" (Anguyo, 2020).

Moreover, food aid typically consists of non-nutrient dense foods such as grain (Ong et al., 2020; Webb et al., 2017). This can cause hidden hunger, as such foods are filling but not sufficiently nutritious, and because their minerals are poorly absorbed (Biesalski, 2017). Cash-transfer programmes are preferable over food aid, where food is available, as they are less costly and less prone to governance challenges (Gentilini, 2016; Ravallion, 2020), but some lockdown measures can affect physical access to food, undermining the effectiveness of cash transfers. Moreover, setting-up ad hoc cash-transfer programmes is a major challenge (Arndt et al., 2020), especially in rural areas. Hammond et al. (2022, p. 9) report an "almost complete absence of direct aid or government support in every location studied" in rural areas in seven countries in the Global South.

The results suggest that all types of lockdown measures affect nutrition security. This highlights the advantage of soft measures, such as face masks and testing, tracing, and isolation strategies to contain the spread of contagious diseases. Some countries, such as South Korea or Vietnam, have been successful in using such strategies, some of which were drawing on community health infrastructure (Gilmore et al., 2020). Yet, using such strategies can be challenging for diseases with long incubation periods or a high proportion of asymptomatic cases, once infections are beyond a certain threshold and when test kits are missing. Moreover, implementing such strategies can be difficult, where public administrations are underfunded or people mistrust the government. Hence, there can be situations where such strategies fail

and lockdowns become necessary. In such cases, there is a large scope for "nutrition-sensitive lockdown policies," which maximize epidemiological benefits and minimize effects on nutrition security.

Lockdown strategies should aim to maximize overall (short-term and long-term) public health by identifying measures that are effective to "flatten the curve" and minimize negative trade-offs with nutrition security. Yet, there are lockdown measures that are particularly effective in flattening the curve but at the same time particularly problematic concerning nutrition sensitivity. This comprises measures related to workplace closures, which are effective to slow contagion but heavily undermine economic access to food, especially for non-farming households living "hand-to-mouth" (Arndt et al., 2020; Devereux et al., 2020; Kansiime et al., 2020; Laborde et al., 2020). Similarly, movement restrictions such as stay-at-home orders are a powerful tool to slow contagion, but they heavily undermine economic and physical access to food. Other measures are less affected by such trade-offs. For example, banning large public events can reduce "super-spreading" (Ebrahim & Memish, 2020; Haug et al., 2020) without compromising nutrition security.

Our article makes the case for nutrition-sensitive lockdown policies, but more research is needed, especially concerning policy recommendations for the future. We focus on the nutrition security of different lockdown measures but do not empirically study the effectiveness of different lockdown measures looking to flattening the curve. Thus, we look only at one side of the above-mentioned trade-off.

Future research should integrate the nutrition security effects of specific lockdown measures into epidemiological models (for an overview of COVID-19 models see Holmdahl & Buckee, 2020). Such models can capture the country-specific trade-offs between different lockdown measures regarding the efficiency to slow contagion (Haug et al., 2020) and nutrition security effects while taking into account health care capacities, the susceptibility of the population to become severely affected as well as the possibility of governments to set up supportive policy to mitigate the impacts of lockdowns measures. This can help to maximize overall (short- and long-term) public health and is important as, in many countries, the nutrition security effects of lockdowns may quickly outweigh epidemiological benefits (see also Meyerowitz-Katz et al., 2021). Coupled models could also help to explore whether circuit-breaker lockdowns—planned and repeated short-term lockdowns—are preferable to prolonged lockdowns (see also Chowdhury et al., 2020).

Future research should disentangle the short- and long-term effects of COVID-19 on nutrition security. Our empirical data ends with the end of the first wave of lockdowns. However, lockdown measures may also undermine the long-term resilience of households to shocks. Households that seemingly cope well with lockdown measures may—below the radar screen of governments—use up all their savings and have to sell their assets, making them vulnerable to a potential second lockdown or other shocks.

As such, how to make food systems more resilient to shocks should be another key research area (Ali et al., 2020; Torero, 2020). This includes resilience for all the above-derived dimensions of nutrition security (production, access, and utilization) and hence explicitly address both the supply and demand side of food systems. On the supply side, policy-makers may improve the level of preparedness for disease outbreaks by having dormant food aid or cash-transfer systems that can be activated in cases of emergencies.

On the supply side, there is a need to design agricultural production and trade in ways that enhance resilience to different types of shocks (such as COVID-19, trade risks, and climate risks). While local food systems may be more resilient to deal with global shocks, global food systems and trade may help to buffer local shocks, suggesting a need to carefully balance the advantages and disadvantages of both. In more resilient food systems, trade-offs between lockdown and nutrition security effects may be reduced.

The article suggests that governments, which face a deadly contagious disease such as COVID-19 and have a population that is susceptible to (hidden) hunger should carefully consider trade-offs between measures for the contagion of the virus and nutrition security. All types of lockdown measures can affect nutrition security as well as other aspects of public health, such as access to vaccinations and health precautions, which can have a large effect on the health of children (Zar et al., 2020). Depending on the situation, test-and-trace strategies may be sufficient (if government capacities allow) and lockdown measures may only be a second-best strategy, but if lockdowns become necessary, they should be designed to be nutrition-sensitive.

WILEY $^{17 \text{ of } 20}$

This in turn may help to ensure compliance with lockdown measures, as food-secure people are more likely to follow the enacted rules (Haug et al., 2020). Imposing strict lockdown measures and hoping to mitigate their negative effects on nutrition security with cash transfers and food aid seems to be a risky strategy given the governance challenges of such programmes, in particular when they are ad hoc responses (Arndt et al., 2020). However, cash transfers and food aid are essential where strict lockdowns have to be imposed. A long-term goal should be better governance to address potential challenges associated with both test-and-trace strategies and food transfer programmes and the design of more resilient food systems.

ACKNOWLEDGMENTS

We are grateful for the financial support from the Program of Accompanying Research for Agricultural Innovation (PARI), which is funded by the German Federal Ministry of Economic Cooperation and Development (BMZ). We are also very grateful for the work of the reviewers and editors involved. Open Access funding enabled and organized by Projekt DEAL.

FUNDING INFORMATION

We received financial support from the Program of Accompanying Research for Agricultural Innovation (PARI), which is funded by the German Federal Ministry of Economic Cooperation and Development (BMZ).

DATA AVAILABILITY STATEMENT

The data that support the findings of this study are available from the corresponding author upon reasonable request.

REFERENCES

- Abay, K. A., Berhane, G., Hoddinott, J. F., & Tafere, K. (2020). COVID-19 and food security in Ethiopia: Do social protection programs protect? (IFPRI Discussion Paper No. 1972). International Food Policy Research Institute. https://doi. org/10.2499/p15738coll2.134146
- Adebayo, B. (2020, April 20). Ghana lifts lockdown, citing improved testing and 'severe' impact on the poor. Cable News Network (CNN). Warner Bros. Discovery. https://edition.cnn.com/2020/04/20/africa/ghana-ends-lockdown-intl/index.html
- Adjognon, G. S., Bloem, J. R., & Sanoh, A. (2020). The coronavirus pandemic and food security: Evidence from West Africa (Policy Research Working Paper No. 9474). World Bank. https://openknowledge.worldbank.org/handle/10986/34775
- Ali, Z., Green, R., Zougmoré, R. B., Mkuhlani, S., Palazzo, A., Prentice, A. M., Haines, A., Dangour, A. D., & Scheelbeek, P. F. D. (2020). Long-term impact of West African food system responses to COVID-19. *Nature Food*, 1(12), 768–770. https:// doi.org/10.1038/s43016-020-00191-8
- Amare, M., Abay, K. A., Tiberti, L., & Chamberlin, J. (2021). COVID-19 and food security: Panel data evidence from Nigeria. Food Policy, 101, 102099. https://doi.org/10.1016/j.foodpol.2021.102099
- Anguyo, I. (2020). The politics of food relief in Uganda's COVID-19 era. LSE. https://blogs.lse.ac.uk/africaatlse/2020/07/01/ politics-food-relief-aid-uganda-covid19/
- Ansah, I. G. K., Gardebroek, C., & Ihle, R. (2019). Resilience and household food security: A review of concepts, methodological approaches and empirical evidence. Food Security, 11, 1187–1203. https://doi.org/10.1007/s12571-019-00968-1
- Arndt, C., Davies, R., Gabriel, S., Harris, L., Makrelov, K., Robinson, S., Levy, S., Simbanegavie, W., van Seventer, D., & Anderson, L. (2020). Covid-19 lockdowns, income distribution, and food security: An analysis for South Africa. *Global Food Security*, 26, 100410. https://doi.org/10.1016/j.gfs.2020.100410
- Ayanlade, A., & Radeny, M. (2020). COVID-19 and food security in Sub-Saharan Africa: Implications of lockdown during agricultural planting seasons. *npj Science of Food*, 4, 13. https://doi.org/10.1038/s41538-020-00073-0
- Ayitey, C. (2020, March 30). Lockdownscare: Panic buyers in Accrameetemptyshelves at supermarkets. My Joy Online. Multimedia Group Limited. https://www.myjoyonline.com/lockdown-scare-panic-buyers-in-accra-meet-empty-shelves-at-supermarkets/
- Berazneva, J., & Lee, D. R. (2013). Explaining the African food riots of 2007–2008: An empirical analysis. *Food Policy*, *39*, 28–39. https://doi.org/10.1016/j.foodpol.2012.12.007
- Barrett, C. B. (2020). Actions now can curb food systems fallout from COVID-19. *Nature Food*, 1(6), 319–320. https://doi. org/10.1038/s43016-020-0085-y
- Bedzrah, E. (2020, April 11). Ghana's pandemic lockdown encounter with the situation on the ground. My Joy Online. https://www.myjoyonline.com/ghanas-pandemic-lockdown-encounter-with-the-situation-on-the-ground/

- Béné, C., Headey, D., Haddad, L., & von Grebmer, K. (2016). Is resilience a useful concept in the context of food security and nutrition programmes? Some conceptual and practical considerations. *Food Security*, 8(1), 123–138. https://doi. org/10.1007/s12571-015-0526-x
- Biesalski, H. K., & Black, R. E. (Eds.). (2016). Hidden hunger: Malnutrition and the first 1,000 days of life: Causes, consequences and solutions. Karger Medical and Scientific Publishers. https://doi.org/10.1159/isbn.978-3-318-05685-3
- Birner, R., Blaschke, N., Bosch, C., Daum, T., Graf, S., Güttler, D., Henia, J., Kariuki, J., Katusiime, R., Seidel, A., Senon, Z.-N., & Woode, G. (2021). 'We would rather die from Covid-19 than from hunger' - exploring lockdown stringencies in five African countries. *Global Food Security*, 31, 100571. https://doi.org/10.1016/j.gfs.2021.100571
- Blaydes, L., & Kayser, M. A. (2011). Counting calories: democracy and distribution in the developing world. International Studies Quarterly, 55(4), 887–908. https://www.jstor.org/stable/41409804, https://doi.org/10.1111/j.1468-2478. 2011.00692.x
- Cara, A. (2020, April 9). Africa must not be 'neglected' in fight against novel coronavirus, officials say. The Globe and Mail Inc. https://www.theglobeandmail.com/world/article-africa-must-not-be-neglected-in-fight-against-novel-coronavirus/
- Ceballos, F., Kannan, S., & Kramer, B. (2020). Impacts of a national lockdown on smallholder farmers' income and food security: Empirical evidence from two states in India. *World Development*, 136, 105069. https://doi.org/10.1016/j. worlddev.2020.105069
- Chowdhury, R., Heng, K., Shawon, M. S. R., Goh, G., Okonofua, D., Ochoa-Rosales, C., ... Shahzad, S. (2020). Dynamic interventions to control COVID-19 pandemic: A multivariate prediction modelling study comparing 16 worldwide countries. *European Journal of Epidemiology*, 35(5), 389–399. https://doi.org/10.1007/s10654-020-00649-w
- Covid-19: How a poultry farmer can stay afloat. (2020, May 16). *Monitor*. Monitor Publications Limited. https://www.monitor. co.ug/Magazines/Farming/Covid19-poultry-farmer-afloat-Kenya-Uganda-market/689860-5554478-11y3fksz/index. html
- Covid-19: How residents of Kabarole donated and rescued vulnerable people. (2020, May 15). *Daily Monitor*. Monitor Publications Limited. https://www.monitor.co.ug/News/National/How-residents-of-Kabarole-donated-and-rescued-vulnerable-people/688334-5553538-s6ljkm/index.html
- Devereux, S., Béné, C., & Hoddinott, J. (2020). Conceptualising COVID-19's impacts on household food security. Food Security, 12(4), 769–772. https://doi.org/10.1007/s12571-020-01085-0
- Ebrahim, S. H., & Memish, Z. A. (2020). COVID-19: Preparing for superspreader potential among Umrah pilgrims to Saudi Arabia. Lancet, 395(10227), e48. https://doi.org/10.1016/s0140-6736(20)30466-9
- FAO. (2008). An introduction to the basic concepts of food security. Food and Agriculture Organization of the United Nations. www.fao.org/3/a-al936e.pdf
- FAO, International Fund for Agricultural Development, United Nations Children's Fund, World Food Programme, & World Health Organization. (2020). The State of Food Security and Nutrition in the World (SOFI): Transforming food systems for affordable healthy diets. *Food and Agriculture Organization of the United Nations*. https://doi.org/10.4060/ca9692en
- Fore, H. H., Dongyu, Q., Beasley, D. M., & Ghebreyesus, T. A. (2020). Child malnutrition and COVID-19: the time to act is now. Lancet, 396(10250), 517–518. https://doi.org/10.1016/S0140-6736(20)31648-2
- Freedom House. (2020). Global Freedom Scores: Countries and Territories. https://freedomhouse.org/countries/ freedom-world/scores
- Gentilini, U. (2016). The revival of the "cash versus food" debate: New evidence for an old quandary? *The World Bank Research* Observer, 31(1), 135–167. https://doi.org/10.1093/wbro/lkv012
- Gerard, F., Imbert, C., & Orkin, K. (2020). Social protection response to the COVID-19 crisis: options for developing countries. Oxford Review of Economic Policy, 36(S1), S281–S296. https://doi.org/10.1093/oxrep/graa026
- Gilmore, B., Ndejjo, R., Tchetchia, A., de Claro, V., Mago, E., Lopes, C., & Bhattacharyya, S. (2020). Community engagement for COVID-19 prevention and control: A rapid evidence synthesis. BMJ Global Health, 5(10), e003188. https://doi. org/10.1136/bmjgh-2020-003188
- Hale, T., Webster, S., Petherick, A., Phillips, T., Kira, B., Petherick, A., Phillips, T., Webster, S., Cameron-Blake, E., Hallas, L., Majumdar, S., & Tatlow, H. (2020). A global panel database of pandemic policies (Oxford COVID-19 Government Response Tracker). Nature Human Behaviour, 5, 529–538. https://doi.org/10.1038/s41562-021-01079-8
- Hammond, J., Siegal, K., Milner, D., Elimu, E., Vail, T., Cathala, P., Gatera, A., Karim, A., Lee, J.-A., Douxchamps, S., Tu, M. T., Ouma, E., Lukuyua, B., Lutakome, P., Leitner, S., Wanyama, I., Thi, T. P., Phuc, P. T. H., Herrero, M., & van Wijk, M. (2022). Perceived effects of COVID-19 restrictions on smallholder farmers: Evidence from seven lower-and middle-income countries. *Agricultural Systems*, 198, 103367. https://doi.org/10.1016/j.agsy.2022.103367
- Harris, J., Depenbusch, L., Pal, A. A., Nair, R. M., & Ramasamy, S. (2020). Food system disruption: Initial livelihood and dietary effects of COVID-19 on vegetable producers in India. *Food Security*, 12(4), 841–851. https://doi.org/10.1007/ s12571-020-01064-5
- Haug, N., Geyrhofer, L., Londei, A., Dervic, E., Desvars-Larrive, A., Loreto, V., Pinior, B., Thurner, S., & Klimek, P. (2020). Ranking the effectiveness of worldwide COVID-19 government interventions. *Nature Human Behaviour*, 4(12), 1303–1312. https://doi.org/10.1101/2020.07.06.20147199

- Headey, D., Heidkamp, R., Osendarp, S., Ruel, M., Scott, N., Black, R., Shekar, M., Bouis, H., Flory, A., Haddad, L., & Walker, N. (2020). Impacts of COVID-19 on childhood malnutrition and nutrition-related mortality. *The Lancet*, 396(10250), 519–521. https://doi.org/10.1016/S0140-6736(20)31647-0
- HLPE. (2017). Nutrition and food systems. A report by the High Level Panel of Experts on Food Security and Nutrition. https:// www.fao.org/3/17846E/i7846e.pdf
- Hirvonen, K., de Brauw, A., & Abate, G. T. (2021). Food consumption and food security during the COVID-19 pandemic in Addis Ababa. American Journal of Agricultural Economics, 103(3), 772–789. https://doi.org/10.1111/ajae.12206
- Holmdahl, I., & Buckee, C. (2020). Wrong but useful—what covid-19 epidemiologic models can and cannot tell us. New England Journal of Medicine, 383(4), 303–305. https://doi.org/10.1056/NEJMp2016822
- Hsiang, S., Allen, D., Annan-Phan, S., Bell, K., Bolliger, I., Chong, T., Druckenmiller, H., Huang, L. H., Hultgren, A., Krasovich, E., Lau, P., Lee, J., Rolf, E., Tseng, J., & Wu, T. (2020). The effect of large-scale anti-contagion policies on the COVID-19 pandemic. *Nature*, 584(7820), 262–267. https://doi.org/10.1038/s41586-020-2404-8
- Huber, C., Finelli, L., & Stevens, W. (2018). The economic and social burden of the 2014 Ebola outbreak in West Africa. The Journal of Infectious Diseases, 218(S5), S698–S704. https://doi.org/10.1093/infdis/jiy213
- Jaacks, L. M., Veluguri, D., Serupally, R., Roy, A., Prabhakaran, P., & Ramanjaneyulu, G. V. (2021). Impact of the COVID-19 pandemic on agricultural production, livelihoods, and food security in India: Baseline results of a phone survey. *Food Security*, 13(5), 1323–1339. https://doi.org/10.1007/s12571-021-01164-w
- Jafri, A., Mathe, N., Aglago, E. K., Konyole, S. O., Ouedraogo, M., Audain, K., Zongo, U., Laar, A. K., Johnson, J., & Sanou, D. (2021). Food availability, accessibility and dietary practices during the COVID-19 pandemic: A multi-country survey. *Public Health Nutrition*, 24(7), 1798–1805. https://doi.org/10.1017/S1368980021000987
- Jahnke, H. E. (1982). Livestock production systems and livestock development in tropical Africa. Kieler Wissenschaftsverlag Vauk.
- Kansiime, M. K., Tambo, J. A., Mugambi, M. I., Bundi, M. M., Kara, A., & Owuor, M. C. (2020). COVID-19 implications on household income and food security in Kenya and Uganda: Findings from a rapid assessment. World Development, 137, 105199. https://doi.org/10.1016/j.worlddev.2020.105199
- Laborde, D., Martin, W., Swinnen, J., & Vos, R. (2020). COVID-19 risks to global food security. Science, 369(6503), 500–502. https://doi.org/10.1126/science.abc4765
- Lau, H., Khosrawipour, V., Kocbach, P., Mikolajczyk, A., Schubert, J., Bania, J., & Khosrawipour, T. (2020). The positive impact of lockdown in Wuhan on containing the COVID-19 outbreak in China. *Journal of Travel Medicine*, 27(3), taaa037. https://doi.org/10.1093/jtm/taaa037
- Leddy, A. M., Weiser, S. D., Palar, K., & Seligman, H. (2020). A conceptual model for understanding the rapid COVID-19-related increase in food insecurity and its impact on health and healthcare. *The American Journal of Clinical Nutrition*, 112(5), 1162–1169. https://doi.org/10.1093/ajcn/nqaa226
- Lockdown has triggered rise in food prices-JCTR. (2020, May 11). *Lusaka Times*. https://www.lusakatimes.com/2020/05/11/ lockdown-has-triggered-rise-in/
- Mahler, G. D., Yonzan, N., Lakner, C., Castaneda, A., & Wu, H. (2021). Updated estimates of the impact of COVID-19 on global poverty: Turning the corner on the pandemic in 2021? World Bank. https://blogs.worldbank.org/opendata/ updated-estimates-impact-covid-19-global-poverty-turning-corner-pandemic-2021
- Mahmud, M., & Riley, E. (2021). Household response to an extreme shock: Evidence on the immediate impact of the Covid-19 lockdown on economic outcomes and well-being in rural Uganda. World Development, 140, 105318. https://doi. org/10.1016/j.worlddev.2020.105318
- Meyerowitz-Katz, G., Bhatt, S., Ratmann, O., Brauner, J. M., Flaxman, S., Mishra, S., Sharma, M., Mindermann, S., Bradley, V., Vollmer, M., Merone, L., & Yamey, G. (2021). Is the cure really worse than the disease? The health impacts of lockdowns during COVID-19. BMJ Global Health, 6(8), e006653. https://doi.org/10.1136/bmjgh-2021-006653
- Minten, B., Mohammed, B., & Tamru, S. (2020). Emerging medium-scale tenant farming, gig economies, and the COVID-19 disruption: The case of commercial vegetable clusters in Ethiopia. The European Journal of Development Research, 32(5), 1402–1429. https://doi.org/10.1057/s41287-020-00315-7
- Moehler, D. C., & Singh, N. (2011). Whose news do you trust? Explaining trust in private versus public media in Africa. Political Research Quarterly, 64(2), 276–292. https://doi.org/10.1177/1065912909349624
- Mwansa, C. (2020, April 18). Covid-19: It is time to temporarily legalise Kachasu. Lusaka. Times. https://www.lusakatimes. com/2020/04/18/covid-19-it-is-time-to-temporarily-legalise-kachasu/
- Ong, M. M., Ong, R. M., Reyes, G. K., & Sumpaico-Tanchanco, L. B. (2020). Addressing the COVID-19 nutrition crisis in vulnerable communities: applying a primary care perspective. *Journal of Primary Care & Community Health*, 11, 2150132720946951. https://doi.org/10.1177/2150132720946951
- Onyango-Obbo, C. (2020, June 10). Let's give the devil his due; our 'rogues' get a B++ on Covid-19. Nation. https://nation. africa/oped/opinion/Let-us-give-the-devil-his-due/440808-5538232-ge94a9/index.html
- Pereira, M., & Oliveira, A. M. (2020). Poverty and food insecurity may increase as the threat of COVID-19 spreads. Public Health Nutrition, 23(17), 3236–3240. https://doi.org/10.1017/S1368980020003493

- Pingali, P., Alinovi, L., & Sutton, J. (2005). Food security in complex emergencies: Enhancing food system resilience. *Disasters*, 29, S5–S24. https://doi.org/10.1111/j.0361-3666.2005.00282.x
- Poelman, M. P., Gillebaart, M., Schlinkert, C., Dijkstra, S. C., Derksen, E., Mensink, F., Hermans, R. C. J., Aardening, P., de Ridder, D., & de Vet, E. (2021). Eating behavior and food purchases during the COVID-19 lockdown: A cross-sectional study among adults in the Netherlands. *Appetite*, 157, 105002. https://doi.org/10.1016/j.appet.2020.105002
- Politicisation of food sharing in lockdown areas unacceptable Group. (2020, April 11). My Joy Online. Multimedia Group Limited. https://www.myjoyonline.com/politicisation-of-food-sharing-in-lockdown-areas-unacceptable-group/
- PRS. (2022). International country risk guide indicator of quality of government. https://www.prsgroup.com/explore-ourproducts/icrg/
- Ravallion, M. (2020, April). Pandemic policies in poor places (CGD Note No. 24). Center for Global Development. https://www.cgdev.org/publication/pandemic-policies-poor-places
- Roberton, T., Carter, E. D., Chou, V. B., Stegmuller, A. R., Jackson, B. D., Tam, Y., Sawadogo-Lewis, T., & Walker, N. (2020). Early estimates of the indirect effects of the COVID-19 pandemic on maternal and child mortality in low-income and middle-income countries: A modelling study. *The Lancet Global Health*, 8(7), e901–e908. https://doi.org/10.1016/ S2214-109X(20)30229-1
- Robinson, E., Boyland, E., Chisholm, A., Harrold, J., Maloney, N. G., Marty, L., Mead, B. R., Noonan, R., & Hardman, C. A. (2021). Obesity, eating behavior and physical activity during COVID-19 lockdown: A study of UK adults. *Appetite*, 156, 104853. https://doi.org/10.1016/j.appet.2020.104853
- Sen, A. (2001). Development as freedom. Alfred A. Knopf.
- Sen, A. (1981). Poverty and famines: An essay on entitlement and deprivation. Oxford University Press. https://doi.org/10. 1093/0198284632.001.0001
- Sharma, S. V., Chuang, R. J., Rushing, M., Naylor, B., Ranjit, N., Pomeroy, M., & Markham, C. (2020). Social determinants of health-related needs during covid-19 among low-income households with children. *Preventing Chronic Disease*, 17. https://doi.org/10.5888/pcd17.200322, E119
- Sinha, D. (2021). Hunger and food security in the times of Covid-19. Journal of Social and Economic Development, 23(2), 320–331. https://doi.org/10.1007/s40847-020-00124-y
- Sinha, I. P., Lee, A. R., Bennett, D., McGeehan, L., Abrams, E. M., Mayell, S. J., Harwood, R., Hawcutt, D. B., Gilchrist, F. J., Auth, M. K. H., Simba, J. M., & Taylor-Robinson, D. C. (2020). Child poverty, food insecurity, and respiratory health during the COVID-19 pandemic. *The Lancet Respiratory Medicine*, 8(8), 762–763. https://doi.org/10.1016/ S2213-2600(20)30280-0
- Thomson, H. (2019). Food and power: Regime type, agricultural policy, and political stability. Cambridge University Press.
- Torero, M. (2020). Without food, there can be no exit from the pandemic. *Nature*, 580, 588–589. https://doi.org/10.1038/ d41586-020-01181-3
- Vendors opt to sleep in markets, walk to work. (2020, March 27). Monitor. Groupe Jeune Afrique. https://www.monitor.co.ug/ uganda/news/national/vendors-opt-to-sleep-in-markets-walk-to-work-1882730
- Vidjingninou, F. (2020, April 9). Au Bénin, Patrice Talon assume l'impossibilité d'un confinement général [In Benin, Patrice Talon assumes the impossibility of a general containment]. Jeune Afrique. https://www.jeuneafrique.com/918313/politique/ au-benin-patrice-talon-assume-limpossibilite-dun-confinement-general/
- von Grebmer, K., Bernstein, J., Mukerji, R., Patterson, F., Wiemers, R., Chéilleachair, N., Foley, C., Gitter, S., Ekstrom, K., & Fritschel, H. (2019). 2019 Global Hunger Index: The challenge of hunger and climate change [Data set]. Welthungerhilfe and Concern Worldwide. https://www.globalhungerindex.org/pdf/en/2019.pdf
- Webb, P., Caiafa, K., Walton, S., & Food Aid Quality Review Group. (2017). Making food aid fit-for-purpose in the 21st century: A review of recent initiatives improving the nutritional quality of foods used in emergency and development programming. *Food and Nutrition Bulletin*, 38(4), 574–584. 1010.1177/0379572117726422
- World Bank. (n.d.). Prevalence of undernourishment (% of the population). https://data.worldbank.org/indicator/SN.ITK. DEFC.ZS
- Wouterse, F., Murphy, S., & Porciello, J. (2020). Social protection to combat hunger. Nature Food, 1(9), 517–518. https://doi. org/10.1038/s43016-020-00144-1
- Zar, H. J., Dawa, J., Fischer, G. B., & Castro-Rodriguez, J. A. (2020). Challenges of COVID-19 in children in low-and middle-income countries. *Paediatric Respiratory Reviews*, 35, 70–74. https://doi.org/10.1016/j.prrv.2020.06.016

How to cite this article: Daum, T., Biesalski, H. K., Blaschke, N., Bosch, C., Güttler, D., Heni, J., Kariuki,

J., Katusiime, R., Seidel, A., Senon, Z.-N., Woode, G., & Birner, R. (2023). Nutrition-sensitive lockdowns:

Conceptual framework and empirical insights from Africa during COVID-19. *Development Policy Review*, 41, e12666. https://doi.org/10.1111/dpr.12666